Author's response to reviews

Title: Associations between social isolation, pro-social behaviour and emotional development in pre-school aged children: a population based survey of kindergarten staff.

Authors:

Louise J Marryat (louise.marryat@glasgow.ac.uk)
Lucy Thompson (lucy.thompson@glasgow.ac.uk)
Helen Minnis (helen.minnis@glasgow.ac.uk)
Philip Wilson (p.wilson@abdn.ac.uk)

Version: 4 Date: 23 September 2014

Author's response to reviews:

Dear Dr Murray,

Please find a new version of the paper entitled 'Associations between social isolation pro-social behaviour and emotional development in pre-school aged children: a population based survey of kindergarten staff'. I have amended the references to match the required style. If you have any other queries, please do not hesitate to get in touch.

Kind Regards,
Louise Marryat (on behalf of all authors)